Author's response to reviews

Title: Bladder augmentation with continent stoma: a surgical option by association between Mitrofanoff's principle and Monti technique.

Authors:

Marcelo F Cassini (marcelo.cassini@globo.com)
Antonio A Rodrigues Jr (aantunesrj@yahoo.com.br)
Silvio Tucci Jr (stuccijr@yahoo.com.br)
Adauto J Cologna (ajcologn@fmrp.usp.br)
Rodolfo B Reis (rbreis@keynet.com.br)
Antonio CP Martins (antonio@capovox.com.br)
Haylton J Suaid (hjsuaid@fmrp.usp.br)

Version: 2 Date: 17 February 2010

Author's response to reviews:

Dear responsible,

I included a title page as the first page of the manuscript, the full names of authors, institutional addresses and respective email addresses for all authors.

The corresponding author was indicated too, as the ethnic origin and nationality of the patient in the case presentation section of the manuscript.

Marcelo F Cassini